Author's response to reviews

Title: Use of GIS to study trauma epidemiology: A pilot study conducted at a trauma care unit.

Authors:

Vaibhav Bagaria (bagariavaibhav@rediffmail.com)
Saurabh Bagaria (saurabhbagaria@gmail.com)

Version: 3 Date: 27 September 2007

Author's response to reviews:

Dear Sir,

Thanks for accepting our manuscript MS: 7313013931404742 titled: Use of GIS to study Trauma Epidemiology: A pilot study conducted at a Trauma Care Unit for publication in your esteemed journal.

We have reformatted the manuscript as per the instructions provided. We look forward to an early publication of the same.

Thanking you in anticipation.

Sincerely yours,

Vaibhav Bagaria
Saurabh Bagaria